Author's response to reviews

Title: A reproducible brain tumour model established from human glioblastoma biopsies.

Authors:

Jian Wang (jian.wang@biomed.uib.no)
Hrvoje Miletic (hrvoje.miletic@biomed.uib.no)
Per Ø Sakariassen (Per.Sakariassen@biomed.uib.no)
Peter C Huszthy (Peter.Huszthy@biomed.uib.no)
Hege Jacobsen (hege.jacobsen@biomed.uib.no)
Narve Brekka (Narve.Brekka@biomed.uib.no)
Xingang Li (lixgang-ql@163.com)
Peng Zhao (drzhaopeng@163.com)
Sverre Mørk (Sverre.Mork@gades.uib.no)
Martha Chekenya (Martha.Chekenya@biomed.uib.no)
Rolf Bjerkvig (rolf.bjerkvig@biomed.uib.no)
Per Ø Enger (per.enger@biomed.uib.no)

Version: 7 Date: 17 December 2009

Author's response to reviews: see over
Dear sirs,

We have made the changes to the format of our manuscript as editor suggested: ‘
1) changed the list of authors in the suggested style. 2) removed the shading within the tables.
3) reduced the size of Figure 1. 4) changed “Material and methods” to “Methods”. 5) have
checked the typography.

Yours sincerely,

Jian Wang, MD&Ph.D
Department of Biomedicine
University of Bergen
Jonas Lies Vei 91
5009 Bergen
Norway
Tel: +47 55586063
Fax: +47 55586360
Email: jian.wang@biomed.uib.no